Analysis of the Determinants of Foreign Direct Investment Flows to the West African Economic and Monetary Union Countries

By

Yélé Maweki Batana PEP, Département d'Économique Université Laval

AERC Research Paper 239 African Economic Research Consortium, Nairobi August 2011 **THIS RESEARCH STUDY** was supported by a grant from the African Economic Research Consortium. The findings, opinions and recommendations are those of the author, however, and do not necessarily reflect the views of the Consortium, its individual members or the AERC Secretariat.

Published by: The African Economic Research Consortium P.O. Box 62882 - City Square Nairobi 00200, Kenya

- Printed by: Modern Lithographic (K) Ltd P.O. Box 52810 - City Square Nairobi 00200, Kenya
- ISBN 978-9966-023-11-7

© 2011, African Economic Research Consortium.

Contents

List of tables List of figures Abstract Acknowledgements 1. Introduction 1 2. Literature review 3 3. Evolution and sector-based distribution of FDI in WAEMU 6 4. The general framework for the study 9 5. Methodology 15 6. Results and discussion 20 7. Conclusion 23 Notes 24 References 25 Appendix 28

List of tables

1.	FDI flows to the WAEMU countries between 1987 and 1998	
	(in US\$ million)	6
2.	FDI flows as a percentage of the GDP and of exports in the WAEMU countries, 1987–1998	7
3.	Evolution of the indexes of risks related to politics and corruption, 1987–1998	9
4.	Some indicators of the education levels in the WAEMU countries	11
5. 6	Key aspects of economic infrastructure in the WAEMU countries The cost of energy telecommunications and transport in the WAEMU	12
	countries in 1995–1996	13
7.	The evolution of the real GDP, the total of exports and imports, and investment in the WAEMU countries between 1991 and 1998	
	(in US\$ million)	13
8.	Inflation rates, tax revenues (in CFAF billion) and foreign debt in the WAEMU countries	14
9.	Results of the Arellano and Bond (1991) GMM estimation, 1970-2002	21
A1:	FDI flows to and investment opportunities in the WAEMU countries, 1996–1998 and 2000–2003	28
A2:	Results of the "within" estimation, 1970–2002	29
A3:	Results of the random effect (RE) estimation, 1970–2002	29
A4:	Results of the Hausman tests, 1970–2002	30
A5:	Matrix of the correlation between consumer prices in France and	
	those in the WAEMU countries between 1970 and 2002	30

List of figures

1.	Evolution of price indexes in France, Cote d'Ivoire	
	and Senegal, 1972-2002	19

Abstract

One of the problems facing sub-Saharan African countries is the low level of domestic investment. And yet the growth theory teaches us that it is impossible to envision development without a considerable accumulation of capital. An important channel through which these countries can solve the problem is to resort to foreign direct investment (FDI), especially since we know the significant role FDI played in the economies of several Asian countries. To date, countries in sub-Saharan Africa have not benefited enough from this type of capital. Several reasons for this exist, and they vary with countries and regions. This study, using dynamic panel data, is an attempt to identify the main determinants of the flows of private foreign investment into countries of the West African Economic and Monetary Union (WAEMU). After a review of the general framework of the study, three estimations were carried out: a "within" estimation, a random effect (RE) estimation, and an estimation using the Arellano and Bond (1991) Generalized Moments Method (GMM). This enables one to get a more effective estimator in cases of dynamic panels. It transpires from the main findings of the research that the rate of domestic investment, literacy, the level of economic openness, and delayed foreign investment are relevant factors that account for foreign investment flows to the WAEMU countries.

Acknowledgements

I would like to express my deep gratitude to the African Economic Research Consortium (AERC) for the financial and logistical support that enabled me to carry out this study. I would also like to thank the thematic group C resource persons for their excellent suggestions. I would equally like to thank the two external reviewers for their sound comments. However, I remain solely responsible for the views and shortcomings of the study.

1. Introduction

The theory of capital teaches us that it is impossible to envision development without a considerable accumulation of capital.¹ If the stock of capital does not indeed reach a certain threshold, the positive effects one would have expected from it would not materialize and the country would fall back into the underdevelopment trap (d'Autume and Michel, 1993). This is the situation that characterizes developing countries and, in particular, those in sub-Saharan Africa where domestic investment remains small (World Bank, 1995a). In such circumstances, achieving sustainable growth requires a substantial flow of foreign capital. Several studies underscore the importance of the role played by foreign direct investment (FDI) in the development of certain countries (World Bank, 1995b; Husain and Wang, 1996). According to Borensztein, Gregorio and Lee (1998), when a reasonable minimum of stock of human capital exists, the contribution of FDI to economic growth is relatively higher than that of domestic investment.

Despite this important role played by FDI, sub-Saharan Africa has not received enough of this type of investment. For example, in 1993, the region managed to attract only US\$700 million or about 1% of the US\$65 billion FDI allocated to all developing countries (Bhattacharya, Montiel and Sharma, 1997). The same trend was observed in 1997, as the relative shares of the developing countries of Asia, Latin America and Africa of the total FDI flows were 22%, 14% and 1% respectively (Mallampally and Sauvant, 1999). According to UNCTAD (2000), despite a slight increase in private capital flows, which rose from US\$8 billion in 1998 to US\$10 billion in 1999, the situation remains unsatisfactory for Africa. Besides, there is an imbalance in the allocation of such capital, as the bulk of it goes to a small number of countries (Morisset, 2000). To attract more FDI, it is imperative to improve the main factors that are likely to influence the choice of possible foreign investors. The aim of this study is to identify all the factors that determine the evolution of the flows of FDI into the countries of the West African Economic and Monetary Union (WAEMU). The countries considered here are Benin, Burkina Faso, Côte d'Ivoire, Mali, Niger, Senegal and Togo.

An analysis of the institutional and socioeconomic framework of the Union's countries reveals certain weaknesses. Some of these are reflected in an ineffective judicial system, relatively high levels of corruption, little effectiveness of the tax legislation, and a weak banking system. Due to relatively low literacy levels, the late 1990s recorded a deterioration of the quality of human capital that was linked to the increase in HIV/AIDS. The economic infrastructure was also found to be little developed, but with significant differences between countries. Based on a dynamic panel model, an econometric analysis has shown that factors such as domestic investment, literacy and the degree of economic

openness played a non-negligible role in attracting FDI flows to the WAEMU region. Moreover, the positive role played by the lagged FDI is an indication that the advantage that is gained from the improvement of incentive factors is a lasting one. It was also observed that the devaluation of the CFA franc in January 1994 was preceded by a flight of capital. However, several factors such as public consumption, gross domestic product, real exchange rate, external debt and political status were not found to be relevant.

The remainder of the paper is structured as follows: Section 2 is a brief review of the literature on the main determinants of FDI. Section 3 presents the evolution and the sector-based distribution of FDI in the WAEMU countries. Section 4 describes the general framework of the study by looking especially at the institutional framework, the state of human resources and economic infrastructure, as well as the macroeconomic framework. Section 5 deals with the research methodology: it deals with the choice of variables, the specification of the model and sources of data. Section 6 presents the main findings and Section 7 gives the conclusion.

2. Literature review

Given that the theory of FDI brings to the fore the importance of the role that this type of investment is likely to play in the development of countries, empirical studies were carried out on the issue, notably concerning Asia (Husain and Wang, 1996). The marked disparities, in space and time, in FDI flows have led economists to reflect upon the deep causes of fluctuations on the world private capital markets. According to Bhattacharya, Montiel and Sharma (1997), the high risk of investing in sub-Saharan Africa would explain why this region has not benefited much from international flows of private capital, of which FDI is a part. It is indeed believed that two types of factors influence capital fluctuations: internal and external factors (Claessens, Dooley and Warner, 1995; Fernández-Arias and Montiel, 1996; Ul Haque, Mathieson and Sharma, 1997). The two causes of fluctuations in external factors generally put forward are: determinants having to do with the economic policy framework and economic determinants (Mallampally and Sauvant, 1999).

The economic policy framework essentially concerns the general regulatory framework for FDI and about trade facilitation measures (UNCTAD, 1998). The principal elements of this regulatory framework are:

- economic, political and social stability;
- the rules governing the inflow and utilization of FDI;
- the standards that are applicable when dealing with foreign subsidiaries;
- operating regulations and market structures;
- international FDI agreements;
- privatization policy;
- trade regulations and how they are in line with FDI regulations; and
- the tax system.

Several studies have sought to capture the link that exists between this type of determinant and FDI inflows. In this regard, Wheeler and Mody (1992) use country risk indexes to demonstrate that there exists a strong correlation between economic and political stability and flows of investment. For Sachs and Sievers (1998), political stability is indeed one of the most important determinants of FDI distribution. According to Singh and Jun (1995, 1996), socio-political instability is a complex phenomenon whose effect is difficult to define, since the estimations carried out to determine the link between this type of instability and FDI most often vary with the political risk indicators used. However, when it is a qualitative index that is used as a measure of the political instability risk, the effect seems significant for the groups receiving large amounts of FDI flows (Singh and Jun, 1995, 1996). The political risk index is indeed statistically

significant at the 1% threshold for countries benefiting from high levels of FDI, while it becomes significant only at the 10% threshold for those receiving small amounts of FDI. The same results will be obtained if one analyses the link between the overall business climate and FDI flows using the qualitative index.

Some experts have mentioned the slow rate of privatization in sub-Saharan Africa, compared with that in Latin America and Eastern Europe, as a reason for limited flows of FDI during the 1980s and the early 1990s. From 1988 to 1994 the sales of state-owned corporations in sub-Saharan African countries amounted to an estimated US\$2.4 billion, compared with US\$63.4 billion in Latin America and US\$16.3 billion in Europe and Central Asia (Bhattacharya Montieland Sharma, 1997). Privatization is an important channel for direct investment and privatization initiatives undertaken in several countries offer great opportunities for FDI (Basu and Srinivasan, 2002). In this connection, Sader (1995) could prove, from a study carried out on developing countries, that the privatization effort undertaken by a given country triggers additional inflows of FDI. Not only does privatization attract FDI through direct sales of state-owned corporations, but it is a sign of improvement in the general business climate to the extent that it reduces state intervention (Pigato, 2000).

Furthermore, Morisset and Pirnia (2000) find that taxation can have a significant impact on flows of foreign investment if the other factors like political and economic stability, infrastructure and transport costs are the same for the countries studied.

As for trade facilitation measures, they have to do with:

- investment promotion;
- incentives to promote investment;
- parasite costs related to corruption, administrative inefficiency, etc.;
- social amenities (bilingual schools, quality of life, etc.);
- post-investment services. (UNCTAD, 1998).

Economic determinants cover areas as varied as the business climate, the search for markets, the search for resources and assets, and the search for efficiency gains. There are several economic factors usually mentioned to account for the evolution of FDI.

The availability of raw materials or an abundant labour force is generally recognized as a factor likely to attract foreign capital (Taylor and Sarno, 1997). According to Morisset (2000), the role of natural resources in the decisions made by multinational companies about investment manifests itself through a sector-based distribution of FDI flows into sub-Saharan Africa. There is generally a more than 60% probability that one FDI dollar that comes to Africa is invested in the natural resources and oil sectors (UNCTAD, 1999b). One other asset would be to have at one's disposal a good level of human capital, which would foster productivity gains.

The rates of wages and investment are potentially important factors accounting for the movement of capital (Brewer, 1991). When such rates are low for wages and high for investment, this means a conducive environment for business. According to Bhattacharya, Montiel and Sharma (1997), the wage bill tends to be higher in sub-Saharan Africa countries than in Asian ones, which might explain the lower level of FDI in sub-Saharan Africa.

The degree of economic openness indicates the level of access to regional and world markets: the bigger the volume of a country's foreign trade, the wider open is its door to portfolio investment (Sader, 1995).

The real interest rate can be an indication of the degree of incentive to place one's

savings in a given country, while the real exchange rate is an indication of the level of the competitiveness of its economy. Public consumption indicates the weight of the government in the economy: the higher it is, the less motivated the investors are to invest in the country (Brewer, 1991; Sader, 1995).

The size of the market and the per capita income can stimulate capital inflows to the country (Mallampally and Sauvant, 1999). It is believed that the low levels of FDI flows to sub-Saharan African countries can be accounted for by the small size of their domestic markets. Per capita income is a measure of the buying power of the domestic market: when it is high, the population is capable of acquiring the goods in the production of which the capital has been invested.

Some empirical studies have been carried out in an attempt to capture the effects of the different factors mentioned above on private capital inflows. For López-Mejía (1999), the considerable inflows of private international capital during the 1990s can be explained by the improvement, at the national level, in the risk-output ratio. This improvement was indeed achieved through:

- the restructuring of foreign debt, which enabled many countries to increase their solvency level;
- structural reforms and confidence building in macroeconomic management, which brought about productivity gains; and,
- the adoption of a fixed exchange rate which, by avoiding the problem related to the exchange rate volatility risk, makes investment more attractive.

From a study carried out on a group of 29 African countries between 1996 and 1997, it was possible to demonstrate the role that the size of the market had played on FDI inflows. The findings highlight a positive correlation between FDI flows and the gross domestic product (GDP), with a correlation coefficient of 0.99 (Morisset, 2000).

Batana (1999) finds that in the case of Togo factors such as rate of investment, degree of economic openness, real interest rate, solvency ratio and public consumption are likely to determine, in the long term, the rate of capital inflows. But in a short-term dynamic only the investment rate and public consumption are found to be relevant.

However, for many observers, the ability of African countries to attract private capital is mainly determined by the availability of natural resources and the size of their domestic market (Morisset, 2000). That is why countries like Nigeria and Angola, despite their political and economic instability, have managed to attract much private capital owing to their oil resources. Nonetheless, Morisset (2000) stresses that African countries can still attract FDI not because of their natural resources but because of their political reforms.

Using the case of MERCOSUR,² a study by Blomström and Kokko (1997) shows that, in the framework of a South–South regional integration, macroeconomic stability seems to be a stronger determinant than regional integration in accounting for FDI inflows. The two authors also emphasize that if integration is well effected, as in the case of MERCOSUR, it can attract considerable investment. However, the fact that the distribution of FDI flows in the WAEMU countries is inequitable is an indication that the way in which regional agreements on investment affect FDI flows depends on the volume of natural resources of each country, the competitiveness of local firms, incentives for investment etc.

That is why it is necessary to study the factors likely to determine the evolution and distribution of FDI flows in countries such as those in the WAEMU zone.

3. Evolution and sector-based distribution of FDI in WAEMU

Evolution of FDI

The information in Table 1 reveals that between 1987 and 1998 FDI flows increased in all the WAEMU countries. Indeed, while the capital flows amounted to US\$111 million on average for the whole Union between 1987 and 1990, they were estimated to be close to US\$459 million between 1995 and 1998. Compared with the share of sub-Saharan African countries put together, that of WAEMU countries rose from an average of 7.6% between 1987 and 1990 to 8.2% for the 1995–1998 period. This increase could be accounted for by the devaluation of the CFA franc. The aim of the devaluation was to make the economies concerned more competitive and, as a result, to lead to an increase in foreign investment in the medium and long term. However, the distribution of FDI flows within WAEMU is characterized by marked imbalance, as Côte d'Ivoire and Senegal get the lion's share, about 80%, of the foreign investment going into the WAEMU zone. However, FDI flows to Mali have increased greatly: from only an average of US\$3 million between 1987 and 1990, they rose to an average of US\$42 million between 1995 and 1998.

Country	Ave	rages by perio	ods	Variations			
	1987–1990	1991–1994	1995–1998	1987–1990	1991–1994	1995–1998	
Benin	1	8	10	1	6	3	
Burkina Faso	4	16	36	3	12	20	
Côte d'Ivoire	51	20	255	12	-32	235	
Mali	3	4	42	1	0	39	
Niger	16	0	0	13	-16	0	
Senegal	24	20	104	31	-4	85	
Togo	12	10	12	8	-2	2	
WĂEMU	111	78	459	69	-38	384	
Sub-Saharan							
Africa	1,455	1,807	5,583	385	352	3,776	

Table 1: FDI flows to the WAEMU countries between 1987 and 1998 (in US\$ million)

Source: Pigato (2000).

This increase could be attributed to the economic, social and political reforms that took place in the country since 1987 as a result of which the Malian economy, which at the time was heavily controlled, was transformed into an increasingly market economy

(World Bank, 1998). There were also notable increases for Côte d'Ivoire, Burkina Faso, Senegal and Benin. The situation in Togo remained relatively the same in the three periods. Niger experienced a notable increase in FDI inflows between 1987 and 1990; the other periods were characterized by negligible inflows.

However, the progression in the increase of FDI inflows was not linear (Table 1). Indeed the WAEMU countries recorded low levels of FDI flows during the 1991–1994 period; most of them (Côte d'Ivoire, Mali, Niger, Senegal, Togo) experienced a negative variation or no FDI flow at all during this period. The biggest losses were recorded by Côte d'Ivoire (-32) and Niger (-16).

The results in Table 2 confirm the increases in FDI flows recorded by Côte d'Ivoire, Burkina Faso, Mali and Senegal: for the four countries, FDI flows as a percentage of GDP and of exports increased between 1987–1990 and 1995–1998. As for Togo, there was no change during the two periods because FDI as a percentage of the GDP represented 0.83% for the former and 0.82 for the latter.

Country		FDI/GDP		FDI/Exports			
	1987–1990	1991–1994	1995–1998	1987–1990	1991–1994	1995–1998	
Benin	0.09	0.50	0.46	0.35	1.75	1.86	
Burkina Faso	0.17	0.76	1.43	1.55	6.38	11.93	
Côte d'Ivoire	0.50	0.16	2.43	1.58	0.61	5.46	
Mali	0.15	0.29	1.62	0.95	0.84	7.27	
Niger	0.66	0.00	0.00	3.92	0.00	0.00	
Senegal	0.45	0.51	2.23	1.92	1.51	6.75	
Togo	0.83	0.66	0.82	2.16	2.42	2.45	

Table 2:	FDI flows	as a percentage	e of the (GDP and	of exports	in the V	VAEMU
	countries	, 1987–1998					

Source: Pigato (2000).

With regard to the origin of FDI, France remains the biggest foreign investor in African countries of the franc zone: its share represents about 40% of the total FDI stock in these countries. Nevertheless, the investment of French firms in this zone is marginal, compared with the total French investment in the world. To illustrate this, in 1968 the franc zone attracted 45% of French FDI; this dropped to between 1% and 3% in the 1990s. Furthermore, the distribution of French FDI in the WAEMU countries is inequitable, as Côte d'Ivoire alone receives more than half of it. Table 2 shows that, in general, the WAEMU countries saw an increase in FDI flows during the 1996–1998 period compared with the 1991–1995 period, with cumulative FDI flows rising from French francs 628 million to French francs 2,686 million for the entire Franc zone. This improvement could be attributed to the active participation of French firms in privatization operations (Banque de France [Bank of France], 1999).

Sector-based distribution of FDI

Generally, information on FDI in Africa is of poor quality (Pigato, 2000). That said, data from a 1999 UNCTAD report (see UNCTAD, 1999a) establishes, with regard to the FDI stock, that in 1997 the primary sector ranked first with US\$1,716 million,

which represents a 53.4% share; then come the manufacturing sector (26.8%), and the service sector (19.8%). However, for FDI flows, of the US\$581 million that African countries received in 1997, 15% went to the primary sector, 32% to the manufacturing sector, and 42% to the service sector. The mines and oil sub-sector represented 9.2% of the FDI received. In the service sector, the bulk of FDI seems to have gone to the finance sub-sector, which was allocated 22.2% of the total amount. This underscores the interest foreign investors have shown in the service sector in the recent years. This interest, in the current international environment characterized by globalization and the development of information and communications technology, seems to provide opportunities for investment in the years to come.

Table A1 in the Appendix shows that between 1996 and 1998 the FDI destined for the WAEMU countries was diversified and covered all the sectors of the economy. Côte d'Ivoire, in particular, received considerable investment in the three sectors: the primary, the manufacturing, and the service sectors.

4. The general framework for the study

The general regulatory framework for FDI

This framework concerns political developments, the judiciary system, legislation and regulations, the financial and banking system, and privatization programmes.

Political developments

The political situation in the WAEMU region was relatively stable during the 1970–2000 period, even though political life in some of the countries was marked by the democratization process in the 1990s. Despite the internal and external tensions experienced by Togo, Burkina Faso and Niger, overall there was political stability in the WAEMU zone.

As seen in Table 3, since 1987 the political situation was relatively more stable in Côte d'Ivoire, with an ICRG (International Country Risk Guide) index higher than 60.³ Senegal ranked second in terms of stability during the same period, with an index of more than 50. Senegal and Côte d'Ivoire were less corrupt than the other countries in the Union with indexes of 50 and 45.8 respectively for the 1995–1998 period. However, while in Senegal the situation did not change throughout that period, in Côte d'Ivoire it deteriorated over time: the corruption risk index was 66.7 between 1991 and 1994. This rise in corruption could be attributed to political tensions and their socioeconomic consequences. For Togo and Niger, indexes between 1995 and 1998 were low (33.3 and 25 respectively), which indicated high risks of corruption.

COUNTRY	Th pol	e ICRG index itics-related	k for risks	The ICRG index for corruption-related risks		
	1987–1990	1991–1994	1995–1998	1987–1990	1991–1994	1995–1998
Benin	-	-	-	-	-	-
Burkina Faso	50.7	47.1	54.4	66.7	54.2	41.7
Côte d'Ivoire	64.5	66.9	63.4	50	66.7	45.8
Mali	37.6	47.1	58.5	22.9	35.4	41.7
Niger	55.3	40.5	45.1	64.6	50	25
Senegal	56.6	53.1	57.1	50	50	50
Togo	45.2	38.6	50.9	33.3	33.3	33.3

Table 3:	Evolution of the index	es of risks	related to	politics and	corruption,
	1987–1998				

Legend: Scale: 0–100; the higher the figure, the lower the risk. Source: Pigato (2000).

The judicial system

The judicial system in all the WAEMU countries seems to be the same; it has a pyramidal structure with high courts at the bottom, then courts of appeal, and, at the top, the Supreme Court.

However, with the exception of Benin, the other countries in the Union should make an extra effort to achieve more independence and transparency for their judiciaries. Benin enjoys relative maturity in matters of independence and transparency of the Judiciary, as has been evidenced by numerous court decisions that went against the country's different political parties.

All the countries have ratified the *Organisation pour l'Harmonisation du Droit des Affaires en Afrique* [Organization for the Harmonization of Business Law in Africa] (OHADA) Treaty, which came into effect in September 1995, with the aim of guaranteeing judicial security within member countries.

Legislation and regulations

Tax laws in the WAEMU countries are characterized by low levels of effectiveness, since government expenditure drastically went up while taxation yields remained weak (Kassé, 1999). According to this author, this weak performance is mainly due to the smallness of the tax base, the inefficiency of the tax collection services and unsuitable tax laws.

As part of the integration measures within WAEMU, a harmonization of the member states' laws was envisaged. In this connection, a legislation harmonization programme was adopted in 1996, with the aim of instituting a legal and tax environment that was conducive to economic activities (Adam, 1999).

The financial and banking system

According to Goreux (1995), the 1990–1991 banking restructuring in the WAEMU countries achieved its main objectives, as the banks have enough liquidity and are thus solvent.

However, overall there seems to be a huge imbalance in the banking system. On average, between 1996 and 1998 Côte d'Ivoire and Senegal had bank statements of accounts totalling about CFAF 2,679 billion which represented two-thirds of the WAEMU statements (Banque de France [Bank of France], 1999). The system was weak because banks were reluctant to fund medium- and long-term investment, in particular investment meant to benefit small- and medium-sized firms (Goreux, 1995). Hernández-Catá et al. (1998) also point out that there was a structural problem in the WAEMU banking system. For example, there was lack of competition illustrated by the big gap between the costs of deposits (2% on average) and lending rates (from 6% to 15%), which resulted in high profitability rates for the banks. According to these authors, such high profitability most often does not encourage banks to change their strategies so as to devise more entrepreneurial lending policies. That is why financial intermediation in those countries should be expanded and deepened and bank guarantees reinforced so as to ensure growth in the medium term.

Privatization programmes

Virtually all the countries in the Union started a policy of withdrawing from the production sector, a withdrawal that was accelerated in the 1990s.

According to data from the World Bank (see World Bank, 2001a), the total amount of the privatization operations in the WAEMU zone between 1990 and 1999 was estimated at about US\$1,114.6 million. Côte d'Ivoire and Senegal shared the bulk of this amount, with US\$597.4 million and US\$410.7 million respectively, representing 90% of the amount for the whole zone. These countries were followed by Benin and Togo, with only US\$39 million and US\$38.1 million respectively. The other countries came last.

The state of human resources

The standards of education in the WAEMU zone are generally low. Only in Togo was more than half the population able to read and write in 1999; the country's rate of literacy was estimated to be 56.4%. Côte d'Ivoire followed (45.7%), Mali (40%), Benin (39%), and Senegal (36.4%). Niger and Burkina Faso had the lowest literacy rates, estimated at 15.3% and 23% respectively. These two rates are definitely inadequate in terms of producing enough skilled workforces for the economy (see Table 4). In comparison, the literacy rate was 84.3% for East Asia and the Pacific and 87.5% for Latin America in 1998.

Togo was also ahead in terms of formal education, with a school enrolment rate from primary to tertiary education of 62%, followed by Benin (45%) and Côte d'Ivoire (38%). This low performance can be attributed in part to the inadequate public resources allocated to the education sector (see Table 4). Furthermore, UNDP data (see PNUD [UNDP], 2000) indicate that between 1995 and 1997 Togo's rate of enrolment in science subjects at the higher education level was relatively low: it was only 11%, while it was 26% in Côte d'Ivoire, 19% in Burkina Faso, and 18% in Mali. Such low rates of enrolment in science subjects may lead to a mismatch between educational programmes and the real needs of the economy.

Country	Adult literacy rates (in % of the population aged 15 and above in 1999)	Gross school enrolment rates (in %) from primary to tertiary education in 1999	Public expenditure on education (in % of GNP) in 1995–1997	
Benin	39	45	3.2	
Burkina Faso	23	23	3.6	
Côte d'Ivoire	45.7	38	5.0	
Mali	39.8	28	2.2	
Niger	15.3	16	2.3	
Senegal	36.4	36	3.7	
Togo	56.3	62	4.5	

Table 4:	Some indicators	of the education	levels in the	WAEMU countries

Source: UNDP (see PNUD, 2001).

In the health sector, the ravages of HIV/AIDS account, to a large extent, for the increase in mortality and the decline in life expectancy in the WAEMU countries in the recent years. UNDP data (see PNUD, 2001) show that Côte d'Ivoire was the WAEMU country most affected by HIV/AIDS in 1999, with an infection rate (the number of people infected by HIV as a percentage of the population aged 15 to 49) of 11%, followed by Burkina Faso (6.44%) and Togo (6%). Senegal and Niger had the lowest rates of 1.77% and 1.35% respectively. It is vital to step up efforts to fight the HIV/AIDS scourge, as it mostly affects the active population and as such weakens human capital. These statistics are higher than those for countries in South Asia (0.5%), Southeast Asia and the Pacific (0.2%) and Latin America (0.5%).

The state of infrastructure

Côte d'Ivoire, Senegal and Togo seem to have the most developed infrastructure networks in the WAEMU zone. Whether we are talking of the number of telephone line subscribers (for every 1,000 people), electricity consumption (in kWh) per head or the ratio of the distance of the road network per inhabitant (1,000 km for 1 million people), the three countries have the highest statistics (see Table 5). As in the case of human resources, Niger seems to have the least developed infrastructure. Overall, the infrastructure network is little developed in the WAEMU countries, especially if compared with other developing countries. For example, in 1998 electricity consumption (in kWh) per head was 320 in South Asia, 670 in Southeast Asia and the Pacific, and 1,452 Latin America (World Bank, 2001b). These figures are much higher than those for the WAEMU countries.

Country	Telephone subscribers (per 1,000 inhabitants) in 1996–1998	Electricity consumption in kWh per head in 1997	Surfaced roads as a % of the total road network in 1998	Road network (1,000 km for 1 million people) in 1996	
Benin	7	48	20	1.2	
Burkina Faso	4	27	16	1.2	
Côte d'Ivoire	12	196	9.7	3.6	
Mali	3	37	12.1	1.5	
Niger	2	38	7.9	1.1	
Senegal	16	135	29.3	1.7	
Тодо	7	97	31.6	1.8	

Table 5:	Key as	pects o	f economic	infrastructure	in the	WAEMU	countries
----------	--------	---------	------------	----------------	--------	-------	-----------

Sources: PNUD [UNDP] (2000); World Bank (2001b).

The information in Table 6 confirms most of the observations made previously, as the cost of various infrastructure services was relatively low in Côte d'Ivoire and Senegal. The weakness of the infrastructure network increases the cost of enterprises and slows down investment.

ANALYSIS OF THE DETERMINANTS OF FOREIGN DIRECT INVESTMENT FLOWS TO THE WAEMU COUNTRIES 13

Country	Electri-	Electri- Diesel Telephone ^a T		Transport ^b	Transport⁵	Transport ^b	
-	city⁵ Industries kWh	oil⁵ CFAF/I	1 r to France	nin to USA	Road per head/km CFAF	Rail per head/km CFAF	by Air ^c CFAF/kg
Benin	41	135	1,760	1,705	47	28.3	2,900
Burkina Faso	51	312	2,262	1,482	42.5	52.5	3,000
Côte d'Ivoire	37.3	270	1,390	870	37.2	31.4	2,900
Mali	55	275	3,060	1,360	43.6	26.3	3,000
Niger	55.4	265	1,800	1,285	50.1	31.4	2,800
Senegal	55.9	300	1,330	800	33.6	26.3	2,200
Тодо	53	195	1,500	857	36.7	42.1	2,800

Table 6:	The cost of energy, telecommunications and transport in the WAEMU
	countries in 1995–1996

^a: 1996, ^b: 1995, ^c: to Paris.

Source: CEFTE (1997)

The macroeconomic framework

Between 1995 and 1998, Côte d'Ivoire and Senegal had the highest GDPs in the WAEMU zone, namely US\$10,987 million and US\$4,875 million respectively, on average. This was also the case for exports and imports which, over the same period, were worth US\$8,638 million for Côte d'Ivoire and US\$3,334 million for Senegal; these two figures correspond to 79% and 68% of their respective GDPs (see Table 7). The relative importance of the economies of the two countries, which can be considered as the WAEMU zone's growth hubs, would to a large extent account for their better performance in terms of FDI. Mali and Burkina Faso also had high investment rates (more than 20% between 1991 and 1998). This investment effort would explain why the FDI flows to the two countries sharply rose between the two periods of 1987–1990 and 1995–1998, from US\$4 million to US\$36 million for Burkina Faso and from US\$3 million to US\$42 million for Mali.

Country	Real GDP		Total of exports and imports (as a % of the GDP)		Gross investment (as a % the GDP)	
	1991–1994	1995–1998	1991–1994	1995–1998	1991–1994	1995–1998
Benin	1,812	2,179	57	59	15	18
Burkina Faso	2,240	2,560	37	39	20	26
Côte d'Ivoire	9,222	1,0987	66	79	9	15
Mali	2,263	2,646	56	56	23	22
Niger	1,798	2,004	39	39	8	10
Senegal	4,223	4,875	71	68	15	18
Togo	1,210	1,432	74	75	13	15

Table 7: The evolution of the real GDP, the total of exports and imports, and investment in the WAEMU countries between 1991 and 1998 (in US\$ million)

Source: World Bank (2000).

The devaluation of the CFA franc in 1994 resulted in price increases. However, the inflation rate remained moderate in the countries in the Union from 1995. Except in Benin (where it was 3.5%) and Burkina Faso (where it was 3.2%), the annual average rate of inflation between 1995 and 1999 remained below 3%, which was the maximum level that WAEMU had set for its member countries. Senegal, Mali and Togo, with respective inflation rates of 1.6%, 2.1% and 2.3%, achieved the best results in curbing inflation during that period (see Table 7). This success in curbing inflation enabled WAEMU to maintain a good proportion of the productivity-price gains that stemmed from the devaluation (Banque de France [Bank of France], 1999).

The information in Table 7 also shows that Côte d'Ivoire and Senegal had the highest tax revenues in the WAEMU zone during the 1996–1999 period, with an average of CFAF1,111.2 billion and CFAF425.1 billion respectively. Niger, as in most cases, had the lowest with only CFAF85.4 billion.

COUNTRY		Δνοτασο	Debt
COONTRI	inflation rate (in %) 1995–1999	tax revenues 1996–1999	ratio/GNP (in %) 1994–1997
Benin	3.5	192.7	85.2
Burkina	3.2	180.3	55.3
Côte d'Ivoire	2.9	1,111.2	206.8
Mali	2.1	207	126.9
Niger	2.7	85.4	87.7
Senegal	1.6	425.1	88.6
Togo	2.3	110.4	120.3

Table 8:	Inflation rates, tax revenues (in C	CFAF billion) and foreig	n debt in the
	WAEMU countries		

Sources: BCEAO (1999); Banque de France (1999).

Between 1994 and 1997 Côte d'Ivoire, Mali and Togo had levels of foreign debt that were higher than their gross national products (GNPs). Their foreign debt ratios vis-à-vis their GDPs were 207%, 127% and 120% respectively. Burkina Faso was the least indebted country, with a debt ratio of about 55%. The implementation of the debt alleviation initiative for the highly indebted poor countries (HIPCs) would alleviate the debt burden on the WAEMU countries, which would put at their disposal additional resources that could be allocated to social sectors. Three countries, namely Burkina Faso, Côte d'Ivoire and Mali, were eligible for the HIPC Initiative.

5. Methodology

The choice of variables

The dependent variable for the model used in this study is the ratio of the amount of flows of FDI over GDP, as is the norm in the literature (Asiedu, 2002).

A review of the theoretical and empirical literature on FDI made it possible to select a number of independent variables likely to influence the foreign investors' decisionmaking process (Edwards, 1990; Sader, 1995; Asiedu, 2002). These variables are described below.

The degree of openness (DO), as measured by the significance of foreign trade, is a factor likely to influence FDI flows, with a positive sign being expected. This is understandable in the sense that most investors prefer to invest in a sector of tradable goods. The degree of openness is measured by the ratio of the total amount of exports and imports to the GDP. Such a ratio is meant to capture in some way the restrictions imposed on international trade. The trade volume can indeed reflect the results achieved by policies aimed at promoting economic openness; these are tariff policies and other incentive measures taken to attract foreign investment.

Domestic investment (INV) is another important factor. It indicates the level of the general business climate, with a positive sign being expected. The increase in domestic investment may indeed mean that the conditions to carry out business activities have improved, which is likely to motivate foreign investors to invest in the country.

Public consumption (GOV) measures the degree of the government's intervention in the economy, with a negative sign being expected because strong government intervention in the economy tends to turn away the private sector and FDI flows.

The real exchange rate (TCR) could prove to be an important factor in the FDI fluctuations on the world market. It is a measure of international competitiveness. However, its impact seems to be ambiguous, as the theoretical and empirical literature on the issue suggests. For instance, while in the short-term real currency depreciation has a negative effect on investment, its long-term effect could prove to be positive (Servers and Solimano, 1992). This thus raises the issue of devaluation, which is carried out with the aim of improving international competitiveness in the medium and long term, while leading to currency depreciation.

The real GDP comes in as a proxy of the size-of-the-market variable, with a positive sign being expected. It indeed measures the effort to increase the size of the market. It is assumed that the bigger the size of the market, the higher the investors' hopes to find enough outlets for their products to achieve economies of scale.

The POL variable was also included in the model used in this study. It indicates the level of liberties in a given country. More specifically, it is a reflection of the level of political rights and that of civil liberties. Since it is meant to measure institutions, this study preferred it to the political risk index (ICRG) used by Singh and Jun (1996), for the simple reason that this index was not available for the period studied.

Another explanatory variable for the model is the literacy rate (ALPH). It measures the quality of labour and can be a strong determinant notably in the area of computer science and the new information and communications technology.

It is often assumed that foreign debt (DET) can also have an impact on foreign investment inflows. An increase in foreign debt can indeed be perceived, to some extent, by investors as a future increase in the taxation rate to finance the servicing of the subsequent debt (Dahl, 2002). That is why its effect is expected to be negative.

The dependent variable delayed for a year (FDI_{t-1}) was included as an explanatory variable in the model, in order to measure the impact of previous investment. This would enable one to come to terms with the idea that an investor decides to invest not because the investment climate is favourable, but rather because previous investment produced satisfactory yields. Singh and Jun (1995) point out that the inclusion of the delayed dependent variable enables one to account for possible autocorrelations of errors, and to indirectly capture the effects of factors omitted from the model, but which may have negatively influenced FDI flows in the past. The expected effect thus seems to be ambiguous.

Specification of the model

The model used in this study is similar to the one used by Sader (1995), but it takes into account the adjustment that investors make over time in order to determine the levels of investment they desire in the face of the constraints which transnational firms undergo (Singh and Jun, 1995). However, the estimation was based on panel data.

Let us posit the following equation:

$$FDIT \quad {}^{d}{}_{it} = \beta_0 + \beta_1 X_{it} + \varepsilon_{it}. \tag{1}$$

Let $FDIT_{it}^{d}$ be the desired stock of FDI in country i and at time t; it is a function of the vector of all the explanatory variables (X_{it}) in country i and at time t, and of a random term ε_{it} . The following equation shows how, for each country i, the real stock of FDI is partially determined based on the difference between the desired stock of FDI and the FDI stock for the previous period. This adjustment is, however, not incompatible with the assumption of rational anticipation that guides the investors' choice about the level of the desired future FDI.

$$FDIT_{it} - = FDIT_{it-1} = \gamma \left(FDIT_{it}^{d} - FDIT_{it-1} \right), \tag{2}$$

ANALYSIS OF THE DETERMINANTS OF FOREIGN DIRECT INVESTMENT FLOWS TO THE WAEMU COUNTRIES 17 where γ represents the speed of adjustment. By combining (1) and (2), and by deriving $FDIT_{ir}$, we get:

$$FDIT_{it} = \gamma B_0 + \gamma B_1 X_{it-1} + (1-\gamma) FDIT_{it-1} + \gamma \varepsilon_{it}$$
(3)

Let us rather consider the flow:

$$FDI_{it} = FDIT_{it} - FDIT_{it-1}, \tag{4}$$

If, in (4) $FDIT_{ii}$ is replaced by its expression in (3), we get:

$$FDI_{it} = \gamma B_0 + \gamma B_1 X_{it-1} + \gamma FDIT_{it-1} + \gamma \varepsilon_{it}$$
(5)

In the final analysis, if the stock of investment (FDIT) is approximated by the FDI flow (FDI) in the right-hand part of the equation, the model to be estimated becomes the following:

$$FDI_{it} = \alpha_0 + \alpha_1 INV_{it} + \alpha_2 GOV_{it} + \alpha_3 POL_{it} + \alpha_4 DO_{it} + \alpha_5 TCR_{it} + \alpha_6 ALPH_{it} + \alpha_7 DET_{it} + \alpha_8 GDP_{it} + \alpha_9 FDI_{it-1} + \alpha_{10} DUM + u_{it}$$
(6)

This approximation can be explained by the choice of the FDI ratio to the GDP. It indeed seemed more reasonable to use the ratio of FDI flows rather than the ratio of the stock investment to GDP. Apart from the variables TCR, GDP, ALPH and POL, all the others are expressed in terms of ratio to GDP. u_{it} designates the random term. In addition, a dummy variable (DUM) was included in the model to capture the anticipation behaviour on the part of investors on the eve of the CFAF devaluation. This variable was thus given the value 1 for the year 1993 and the value 0 for all the other years.

The model was estimated using three procedures: the individual-random-effects estimator (RE), the "within" estimator and the Generalized Moments Method (GMM) estimator used by Arellano and Bond (1991). The first procedure was turned into a simple "pooling" estimation which applied the method of ordinary least squares (OLS) to Equation 6 because of the limited number of countries which could not make the "between" estimation possible. With the second procedure, it was a question of eliminating individual effects. Indeed, by positing $u_{it} = \beta_i + v_{it}$, where β_i is the individual effect and v_{it} the stochastic component which is assumed not to be correlated with the explanatory variable, the "within" estimator was obtained by applying OLS to the following equation:

RESEARCH PAPER 239

$$\left(IDE_{ii} - \overline{IDE}_{ii}\right) = \phi\left(X_{ii} - \overline{X}_{ii}\right) + \gamma\left(IDE_{ii-1} - \overline{IDE}_{ii-1}\right) + u_{ii} - u_{ii}$$
⁽⁷⁾

where \overline{IDE}_{it} and \overline{X}_{it} are the means of the dependent variable and the independent variables. Taking into account the deviations from the individual means enabled the individual effects, assumed to be correlated with the explanatory variables, to be cancelled out, which should produce an unbiased estimator. But the two estimators (RE and "within") are biased and non-convergent because of the autoregressive nature of the model. That is how the following relation was obtained:

$$E\left(FDI_{it-1}\beta_i\right) \neq 0 \tag{8}$$

Moreover, nothing rules out the existence of a correlation between the other explanatory variables and the fixed individual effect β_i . In the case of dynamic panels, the GMM estimator of Arellano and Bond (1991) is most often used in order to deal with the bias of auto-regressiveness and simultaneity. This estimator is used to manipulate the first differences of variables by using the levels of these variables delayed by at least two periods on two assumptions: that variables are predetermined and that error terms are not auto-correlated. That is how convergent estimators are obtained.

Sources of data

All the data come from the databases of the World Bank (World Tables, Global Development Finance and World Development Indicators) and the IMF (International Financial Statistics) and Freedom House. In all, seven countries were covered by this study, with data spanning the 1972–2002 period. The data bore especially on FDI ratios, the investment rate, public consumption and foreign debt ratios, the literacy rate, and the degree of economic openness as measured by the ratio of exports to the GDP, and the real GPD. There are also data on prices and nominal exchange rates which were used to estimate the real exchange rate. This rate was estimated using the following formula:

$$TCR = TCN \times \frac{P^*}{P}$$

where TCN, represents the nominal exchange rate, P* the consumer price index in France, and P the consumer price index in the WAEMU countries. France was chosen because of its status as the majority investor. Figure 1 shows the link between the evolution of prices in France, Côte d'Ivoire and Senegal from 1970 to 2002. Table A5 in the Appendix also shows that prices in the WAEMU countries were strongly correlated with those in France.

18

ANALYSIS OF THE DETERMINANTS OF FOREIGN DIRECT INVESTMENT FLOWS TO THE WAEMU COUNTRIES 19





The data on levels of liberties (POL) in the different countries were obtained from Freedom House. Three levels were considered: absence of liberty, existence of partial liberty, and advanced state of liberty. The three were represented by the numbers 1, 2, and 3 respectively. A positive correlation was thus expected between the variable POL and the FDI evolution.

6. Results and discussion

The results of the "within" and RE estimations, and of the Hausman test, are presented in Tables A2 to A4 in the Appendix. Generally, most of the explanatory factors which the model used in this study took into account did not turn out to be relevant. The "within" and RE estimations produced quite similar results in terms of the main relevant factors. The Hausman test did not lead to a rejection of the null hypothesis, that of the lack of systematic difference between the coefficients of the two estimators. Even though it was not possible to systematically reject the exogeneity of individual effects, which amounts to favouring the RE estimation, it nevertheless seemed reasonable to resolve the auto-regressiveness bias that is inherent in the inclusion, in the model, of the lagged FDI. That is why the Arellano and Bond (1991) GMM estimator was brought in, as it gives robust results (see Table 9). The estimation first considered a wider model where the delays in explanatory variables were also included. A tighter model was thus obtained that excluded the least relevant variables.

This third type of estimation confirms the relevance of certain variables which were already shown by the other two estimations. However, the results are closer to those obtained with the "within" estimation. The Sargan tests of overidentification restrictions did not lead to the rejection of the null hypothesis, which means that the moment conditions were valid. Moreover, an absence of auto-correlation of errors was obtained. Indeed, according to the test proposed by Arellano and Bond (1991), the u_{it} distribution is not auto-correlated when the first-order auto-correlation is negative and significant while the second-order auto-correlation is not significant, which was the result obtained in this case. The results were also robust for the various changes in the number of delays taken into account for the variables serving as instruments.

For the period studied, the results in Table 9 show that the variables that turned out to be relevant in accounting for the foreign direct investment flows to the WAEMU countries are: investment, the literacy rate, the degree of openness, the binary variable, and the lagged FDI.

The domestic investment (INV) had a positive and significant effect on FDI flows in spite of the small amount of money globally invested in the countries in the WAEMU zone. In theory, investment can only produce the expected beneficial effects when it reaches a given optimal level. Yet, in the WAEMU zone, except for Burkina Faso and Mali which achieved significant levels of investment over the 1994–1998 period, the other countries recorded ratios of gross investment to GDP that were lower than 20%. According to Mlambo and Oshikoya (2001), the low levels of investments increase their vulnerability. However, the positive sign obtained shows that the level of investment

ANALYSIS OF THE DETERMINANTS OF FOREIGN DIRECT INVESTMENT FLOWS TO THE WAEMU COUNTRIES 21

can turn out to be an important signal of the state of the business climate in a country. When conditions are created to attract domestic investors, this also attracts more foreign investors.

The contribution of the labour factor to the production process is appreciated through the productivity of this factor. After all, it is axiomatic that a high level of education will enable a country's working population to be more competitive internationally. Moreover, a higher level of human capital enables individuals to better adjust to new production and export structures. In this study, the rate of literacy (ALPH) was found to be relevant; its sign was positive, an indication of the role that it can play in attracting foreign capital into the WAEMU countries.

As revealed by the literature, when the ratio of foreign trade to GDP is high, this is an indication of weak trade barriers, which attract FDI. In the WAEMU countries, the relationship between this variable and FDI was found to be statistically significant, in agreement with Asiedu's (2002) finding that the degree of openness of the economy was a determinant in FDI flows to sub-Saharan African countries. This finding shows that in spite of their marginalization in terms of the world trade, their opening the economy attracted foreign investors. Indeed, a good part of the period covered by this study saw the liberalization of economic activities that went hand in hand with the progressive removal of trade barriers.

Variables	Coefficients	Standard deviations	Z	P > Z
Domestic investment (INV)	0.05**	0.021	2.59	0.010
Public consumption (GOV) Literacy	-0.06	0.042	-1.33	0.184
(ALPH)	0.08***	0.030	2.73	0.006
Degree of openness (DO)	0.04*	0.021	1.74	0.082
Gross domestic product (GDP)	-0.003	0.003	-1.07	0.286
Real exchange rate (TCR) Foreign debt	-0.0006	0.001	-0.57	0.566
(DET) Political status	-0.002	0.005	-0.35	0.728
(POL)	0.079	0.174	0.45	0.650
Binary variable (DUM)				
(1 for 1993 and 0 other years)	-1.15**	0.526	-2.18	0.029
Lagged FDI	0.25***	0.071	3.52	0.000
	Number of obse Number of grou	ervations = 210 ps = 7		
	Sargan's test ch	$i^{2}(410) = 204.85$	Prob >	chi ² =1.00
	Autocorrelation	test AR(1) Z = -7,96	Prob >	Z = 0.000
	Autocorrelation	test AR(2) Z = 1,90	Prob >	Z = 0.368

Table 9: Results of the Arellano and Bond (1991) GMM estimation, 1970-2002

*, ** and *** mean that the coefficients are significant at the respective threshold levels of 10%, 5% and 1% .

Regarding the dummy (DUM) variable a negative sign was obtained which was significant with all the specified models. This finding suggests that some foreign investors held back their investment on the eve of the devaluation. Since the dependent variable was the net FDI, this finding can equally mean that a certain amount of capital was brought back to the country of origin by investors who were concerned about preserving the monetary value of their assets.

Finally, it appears that there was a non-stop flow of FDI, since the results that the FDI of the previous year had a positive effect on the following one. The theory of investment provides justification for such a situation. Singh and Jun (1996) obtained the same result. This means that other factors not taken into account in a model whose effects can be observed in the short or medium term might influence the results and that foreign investment can itself constitute an incentive for further investment, if it generates positive returns or if it is invested in sectors that improve the economic environment of the country (like infrastructure, education etc.).

Contrary to expectations, potentially significant factors like public consumption, the real exchange rate, the institutional variable, the size of the market, and foreign debt did not turn out to be significant.

The coefficient of public consumption (GOV) was found to be negative but not significant. Contrary to the idea that a high level of public consumption results in the reduction of public investment, which happens to be complementary to private investment, be it domestic or foreign (Greene and Villanueva, 1991; Servers and Solimano, 1993; Mlambo and Elhiraika, 1997), the results of this study did not highlight such a negative effect. Moreover, in theory, public consumption can also have a positive effect on FDI flows, as it constitutes potential demand that is likely to stimulate the supply of goods and services.

The same applies to the real exchange rate (TCR). Although, in theory, a real depreciation (or appreciation) of the real exchange rate will result in the increase (or decrease) of FDI flows, it seems that in the case of WAEMU foreign investors did not respond to this factor based on the evolution of consumer prices in France. In general, while depreciation can render production more expensive due to the import of certain inputs, it improves the international competitiveness of the economy. This is plausible in the sector of export-oriented high-intensity labour activities. Empirical research has indeed shown that the relative cost of labour is significantly correlated with foreign investment, particularly in high-intensity labour industries and export-subsidies industries (Marr, 1997). When the bulk of FDI is destined for the privatization of big public corporations that supply domestic services, it is probable that depreciation will not produce the expected effect.

With regard to the institutional variable (POL) that indicates the level of liberties the coefficient was positive, although not significant. The same finding was observed by Morisset (2000), who instead used the ICRG variable as an alternative in a study that covered 29 sub-Saharan Africa countries, including five covered by this study. Even if investors took into account the political situation, especially in countries that are not rich in mineral resources, the relative stability of the WAEMU countries and the weak variability in the indicator used did not bear this concern out.

Finally, the fact that the real GDP was not found to be a relevant factor could be explained by the relative small size of the market in the WAEMU countries. As for the fact that foreign debt was not found to be significant either, that could be explained by the fact that, relatively, most of these countries are not heavily indebted. Besides, the debt alleviation schemes such as the HIPC Initiative, for which certain countries like Burkina Faso, Mali and Côte d'Ivoire are eligible, are of the kind to reassure foreign investors.

7. Conclusion

T is acknowledged that although FDI can be a windfall for economic development, the WAEMU countries do not receive enough of it. This study has identified a number of factors that can account for this phenomenon. The review of the general framework for the study helped to underscore the weakness of the general business climate in the WAEMU zone, characterized especially by low levels of human capital, low levels of domestic investment, the small size of the market, and an infrastructure sector that is small compared to that in Asian and Latin American countries.

From the main econometric results obtained in this study it transpires that domestic investment, literacy, degree of openness, and delayed foreign investment are determinants of FDI flows. On the whole, the level of real GDP seems to be too low for its evolution to reflect, from the investors' point of view, a considerable increase in the size of the market. For its part, foreign debt does not seem to have reached the level at which it can inspire an expression of expectations about a future increase in taxation rates. Another explanation could be the weak variability of factors which did not make it possible to establish a correlation with the variations in FDI flows.

From the discussion above, with levels of domestic investment being low, incentive measures should be taken to raise them, which is likely to affect, directly and indirectly, flows of foreign investment into the WAEMU countries. The privatization policy should be pursued to reduce public consumption and increase expenditure on priority infrastructure. Such a measure should also curb the countries' rate of borrowing, which would enable them to meet the WAEMU convergence criterion that requires the ratio of the outstanding domestic and foreign debt in relation to GDP to be lower than 70%.

Finally, the results of the estimations carried out in this study suggested that a good number of explanatory variables were not taken into account in the model used. Because of the small number of countries that make up WAEMU, it was not possible to include factors that are potentially significant, such as the HIV/AIDS pandemic which is spreading at an alarming rate in these countries. It is not unreasonable to think that the extent of the spread of this pandemic is likely to influence foreign investors' decisions and thus make them hold their investment back. A study based on a bigger sample of countries could enable the tackling of this other dimension of international capital flows. Much still needs to be done to enable a deeper understanding of the evolution of direct investment in Africa, since the lack of correlation, as found in this study, with most of the factors is an indication that there are factors capable of slowing down investment and which are capable of weakening the effects of macroeconomic variables.

Notes

- 1. This is known in the literature on development as the "big push theory", a big push without which developing countries could not achieve their economic take-off (Nurkse, 1953).
- 2. Southern Common Market in South America.
- 3. The International Country Risk Guide (ICRG) index enables one to assess the risks related to the climate of investment and trade. It is measured on a 0-to-100 scale where 0 corresponds to the maximum risk and 100 to the minimum.

References

- Adam, M.K. 1999. "Harmonisation fiscale et l'intégration économique de l'Afrique de l'Ouest : Les TVA harmonisées de la CEDEAO et de l'UEMOA". Le point de l'intégration. A CEA/ SRDC biannual information bulletin, United Nations.
- Arellano, M. and S. Bond. 1991. "Some tests of specification for panel data: Monte Carlo evidence and an application to employment equations". *Review of Economic Studies*, 58: 277–97.
- Asiedu, E. 2002. "On the determinants of foreign direct investment to developing countries: Is Africa different?" *World Development*, 30(1): 107–19.
- D'Autume, A. and P. Michel. 1993. "Hystérésis et piège du sous-développement dans un modèle de croissance endogène". *Revue économique*, 44(2): 431–50.
- Banque de France [Bank of France]. 1999. La zone franc: Rapport annuel 1998.
- Basu, A. and K. Srinivasan. 2002. "Foreign direct investment in Africa: Some case studies". IMF Working Paper. The International Monetary Fund, Washington, D.C., February.
- Batana, Y.M. 1999. "Les déterminants des flux de capitaux privés internationaux au Togo". Les GREAT Cahiers, 2: 22–30.
- BCEAO. 1999. Rapport annuel 1999. Dakar, Senegal: Central Bank of West African States.
- Bhattacharya, A., P.J. Montiel and S. Sharma. 1997. "Afrique subsaharienne: Comment attirer davantage de capitaux privés?". *Finances et Développement*, 34(2): 3–6.
- Blomström, M. and A. Kokko. 1997. "Regional integration and foreign direct investment: A conceptual framework and three cases". Policy Research Working Paper No. 1750. The World Bank, Washington, D.C.
- Borensztein, E., J. de Gregorio and J.-W. Lee. 1998. "How does foreign direct investment affect economic growth?" *Journal of International Economics*, 45: 115–35.
- Brewer, L.T. 1991. "Foreign direct investment in developing countries: Patterns, policies, and prospects". Policy Research Working Paper No. 712. The World Bank, Washington, D.C.
- CEFTE [Centre d'Etudes et de Formation aux Technologies Economiques]. 1997. "Programme d'Appui Régional à l'Intégration des pays de l'UEMOA (PARI)". UEMOA, Final Report, November.
- Claessens, S., M.P. Dooley and A. Warner. 1995. "Portfolio capital flows: Hot or cold?". *The World Bank Economic Review*, 9(1): 153–74.
- Dahl, J. 2002. "Incentives for foreign direct investment: The case of SADC in the 1990s". NEPRU Working Paper No. 81. The Namibian Economic Policy Research Unit, Windhoek, Namibia.
- Edwards, S. 1990. "Capital flows, foreign direct investment, and debt-equity swaps in developing countries". NBER Working Paper No. 3497. National Bureau of Economic Research, Cambridge, Massachusetts, October.
- Fernández-Arias, E. and P.J. Montiel. 1996. "The surge in capital inflow to developing countries: An analytical overview". *The World Bank Economic Review*, 10(1): 51-77.
- Goreux, M.L. 1995. *La dévaluation du franc CFA : Un premier bilan en décembre 1995*. Document de travail, Banque Mondiale.

- Greene, J. and D. Villanueva. 1991. "Private investment in developing countries: An empirical analysis". *IMF Staff Papers*, 38(1): 33–58.
- Hernández-Catá, E., C.A. François, P. Masson, P. Bouvier, P. Peroz, D. Desruelle, and A. Vamvadikidis. 1998. "The West African Economic and Monetary Union: Recent developments and policy issues". Occasional Paper No. 170. International Monetary Fund, Washington, D.C.
- Husain, I. and Y. Wang. 1996. "Investissement direct étranger en Chine et croissance économique". *Revue d'Economie du Développement*, 1, (2).
- Kassé, M. 1999. "Le rôle intégrateur de la fiscalité dans les pays de l'Afrique de l'Ouest. *Le point de l'intégration*. A CEA/SRDC biannual information bulletin. United Nations, December.
- López-Mejía, A. 1999. "Flux massifs de capitaux: Leurs causes, leurs conséquences et la manière de les gérer." *Finances et Développement*, 36(3): 28–31.
- Mallampally, P. and K.P. Sauvant. 1999. "L'investissement direct étranger dans les pays en développement". *Finances et Développement*, 36(1): 34–37.
- Marr, A. 1997. "Foreign direct investment flows to low-income countries: A review of the evidence". ODI Briefing Paper No. 3. Overseas Development Institute, London, September.
- Mlambo, K. and A.B. Elhiraika. 1997. "Macroeconomic policies and private saving and investment in SADC countries". Economic Research Papers No. 33. Abidjan: African Development Bank.
- Mlambo, K. and W.T. Oshikoya. 2001. "Macroeconomic factors and investment in Africa". *Journal of African Economies*, 10 (AERC Supplement 2): 12–47.
- Morisset, J. 2000. "Foreign direct investment in Africa: Policies also matter". World Bank Policy Research Working Paper No. 2481. The World Bank, Washington, D.C.
- Morisset, J. and N. Pirnia. 2000. "How tax policy and incentives affect foreign direct investment: A review". World Bank Policy Research Working Paper No. 2509. The World Bank, Washington, D.C., December.
- Nurkse, R. 1953. *Problems of Capital Formation in Underdeveloped Countries*. Oxford: Basil Blackwell.
- Pigato, M. 2000. "Foreign direct investment in Africa: Old tales and new evidence". Mimeo. The World Bank, Washington, D.C.
- PNUD [UNDP]. 2001. Rapport mondial sur le développement humain 2001: Mettre les nouvelles technologies au service du développement humain. New York: United Nations Development Programme and Brussels: DeBoeck University.
- PNUD [UNDP]. 2000. Rapport mondial sur le développement humain 2000: Droits de l'homme et développement humain au service de la liberté et de la solidarité. New York: United Nations Development Programme and Brussels: DeBoeck University.
- Sachs, J. and S. Sievers. 1998. "Foreign direct investment in Africa". In World Economic Forum, *The Africa Competitiveness Report 1998*. Geneva: World Economic Forum.
- Sader, F. 1995. "La privatisation des entreprises publiques et l'investissement étranger dans les pays en développement, 1988–93". FIAS, No. 5F. Washington D.C.: The World Bank.
- Servers, L. and A. Solimano. 1992. "Private investment and macroeconomic adjustment: A survey". *The World Bank Research Observer*, 7(1): 1 16.
- Singh, H. and K.W. Jun. 1996. "The determinants of foreign direct investment in developing countries". *Transnational Corporations*, 5(2): 67–105.
- Singh, H. and K.W. Jun. 1995. "Some new evidence on determinants of foreign direct investment in developing countries". World Bank Policy Research Working Paper No. 1531. The World Bank, Washington, D.C., November.
- Taylor, P.M. and L. Sarno. 1997. "Capital flows to developing countries: Long and short-term determinants". *The World Bank Economic Review*, 11(3): 451–70.
- Ul Haque, N. D. Mathieson and S. Sharma. 1997. "Causes of capital inflows and policy responses to them". *Finance and Development*, 34(1): 3-6.

26

ANALYSIS OF THE DETERMINANTS OF FOREIGN DIRECT INVESTMENT FLOWS TO THE WAEMU COUNTRIES 27

- UNCTAD. 2000. World Investment Report: Cross-border Mergers and Acquisitions and Development. Overview. New York and Geneva: United Nations Conference on Trade and Development.
- UNCTAD. 1999a. Foreign Direct Investment in Africa: Performance and Potential. New York and Geneva: United Nations Conference on Trade and Development.
- UNCTAD. 1999b. World Investment Report: Foreign Direct Investment and the Challenge of Development. New York and Geneva: United Nations Conference on Trade and Development.
- UNCTAD. 1998. World Investment Report 1998: Trends and Determinants. New York and Geneva: United Nations Conference on Trade and Development.
- Wheeler, D. and A. Mody. 1992. "International investment location decisions: The case of U.S. firms". *Journal of International Economics*, 33: 57–76.
- World Bank. 2001a. World Development Indicators 2001. Washington D.C.: The World Bank.
- World Bank. 2001b. World Development Report 2000/2001: Attacking Poverty. Washington, D.C.: Oxford University Press and The World Bank.

World Bank. 2000. World Development Indicators 2000. Washington D.C.: The World Bank.

- World Bank. 1998. "The heavily indebted poor countries debt initiative: Sustainable debt for sustainable development". Press Release No. 99/1941. The World Bank, Washington, D.C.
- World Bank. 1995a. Un continent en transition: l'Afrique subsaharienne au milieu des années 90, November. Washington, D.C.: The World Bank.
- World Bank. 1995b. *Development of the Private Sector in Low Income Countries*. Washington, D.C.: The World Bank.

Appendix

Sector	FDI flows from	Investment		
	Significant	Not significant	None	opportunities between 2000 and 2003
Primary				
Agriculture	Côte d'Ivoire Mali	Burkina Togo	Niger	Mali, Niger Togo
Fishery Forestry	Côte d'Ivoire	Тодо	Niger Togo	Togo Niger Togo
Mines and extraction	Burkina, Togo Côte d'Ivoire Mali, Niger	-	-	Côte d'Ivoire Niger
Oil, gas and their by-products Manufacturing	Burkina Côte d'Ivoire	Niger	Тодо	Niger
Foods	Côte d'Ivoire	Burkina	Niger	Côte d'Ivoire
and drinks	Togo	Mali	Niger	Mali, Niger, Senegal, Togo
Tobacco	Burkina	Côte d'Ivoire	Niger, Togo	-
Textiles, leather and	Mali	Burkina, Togo		Mali, Niger
clothing	Niger	Côte d'Ivoire	-	Senegal
Chemical and pharmaceutical	Côte d'Ivoire	Mali, Niger Togo	-	Senegal
Metals and metallurgical industry	Burkina	-	Côte d'Ivoire Niger, Togo	-
Mechanical and electrical equipment	-	Côte d'Ivoire	Burkina, Niger, Togo	Senegal
Motor industry	-	-	Niaer. Togo	-
Mineral and non-metal products	Burkina Côte d'Ivoire	Niger Togo	-	Niger
Telecommunications	Burkina, Mali	Τοσο	Niger	Mali, Niger
	Côte d'Ivoire	- 3 -		Senegal, Togo

Table A1: FDI flows to and investment opportunities in the WAEMU countries, 1996–1998 and 2000–2003

continued next page

ANALYSIS OF THE DETERMINANTS OF FOREIGN DIRECT INVESTMENT FLOWS TO THE WAEMU COUNTRIES 29

Sector	FDI flows from 19	Investment		
	Significant	Not significant	None	between 2000 and 2003
Services				
Finance and insurance	Côte d'Ivoire	Burkina, Mali, Togo	Niger	-
Transport and storage	-	Burkina, Mali, Côte d'Ivoire Togo	Niger	Côte d'Ivoire
Tourism	Burkina Côte d'Ivoire	Mali Togo	Niger	Côte d'Ivoire Mali, Niger

Table A1 Continued

Source: UNCTAD (1999b).

Table A2: Results of the "within" estimation, 1970–2002

Variables	Coefficients	Standard deviation	n Z	P > Z
Domestic investment (INV)	0.05**	0.021	2.56	0.011
Public consumption (GOV)	-0.06	0.043	-1.31	0.191
Literacy (ALPH)	0.08***	0.031	2.70	0.008
Degree of openness (DO)	0.04*	0.021	1.72	0.087
Gross domestic product (GDP)	-0.003	0.003	-1.06	0.293
Real exchange rate (TCR)	-0.001	0.001	-0.57	0.571
Foreign debt (DEX)	-0.002	0.005	-0.34	0.732
Political status				
(POL)	0.08	0.176	0.45	0.655
Binary variable				
(DUM)				
(1 for 1993 and 0 otherwise)	-1.15**	0.532	-2.15	0.032
Lagged FDI	0.25***	0.072	3.48	0.001
	Number of ob Number of gro	servations = 217 pups = 7		
	F(10.200)	= 8.66	Prob > F	= 0.00

* = coefficient significant at the 10% level; ** = coefficient significant at the 5% level; *** = coefficient significant at the 1% level.

Variables	Coefficients	Standard deviation	Z	P > Z
Domestic investment (INV)	0.04**	0.018	2.38	0.018
Public consumption (GOV)	-0.10***	0.036	-2.67	0.008
Literacy (ALPH)	0.02*	0.013	1.65	0.100
Degree of openness (DO)	0.02	0.012	1.46	0.146
Gross domestic product (GDP)	-0.001	0.001	-1.13	0.257
Real exchange rate (TCR)	0.0001	0.001	0.10	0.922
Foreign debt (DEX)	0.003	0.003	0.86	0.390
Political status				
(POL)	0.20	0.147	1.33	0.182
			continue	d next page

Variables	Coefficients Standa	ard deviation	Z	P > Z
Binary variable (DUM)				
(1 for 1993 and 0 otherwise)	-1.09**	0.531	-2.06	0.039
Lagged FDI	0.29***	0.069	4.25	0.000
	Number of observatio Number of groups Wald chi²(10)	ns = 217 = 7 = 97.42	Prob > chi²	= 0.00

Table A3 Continued

* = coefficient significant at the 10% level; ** = coefficient significant at the 5% level; *** = coefficient significant at the 1% level.

Table A4: Results of the Hausman tests, 1970–2002

	The "within" coefficient Convergent (A)	The RE coefficient Efficient (B)	Difference (A-B)	S.E.
Domestic investment (INV)	0.05	0.04	0.013	0.012
Public consumption (GOV)	-0.06	-0.10	0.039	0.024
Literacy (ALPH)	0.08	0.02	0.062	0.028
Degree of openness (DO)	0.04	0.02	0.019	0.017
Gross domestic product (GDP)	-0.003	-0.001	-0.002	0.003
Real exchange rate (TCR)	-0.001	0.0001	-0.001	0.001
Foreign debt (DEX)	-0.002	0.003	-0.004	0.003
Political status				
(POL)	0.08	0.20	-0.117	0.098
Binary variable (DUM)				
(1 for 1993 and 0 otherwise)	-1.15	-1.09	-0.052	0.040
Lagged FDI	0.25	0.29	-0.044	0.020

A is convergent under the null hypothesis Ho and the alternative hypothesis Ha

B is non-convergent under the hypothesis Ha and is efficient under the null hypothesis Ho.

Test: Ho: the difference between the coefficients is not systematic

 $Prob > chi^2 = 0.68$

Table A5: Matrix of the correlation between consumer prices in France and those in the WAEMU countries between 1970 and 2002

	Niger	Mali	France	CI	Burkina	Benin	Togo	Senegal
Niger	1.00	0.94	0.92	0.93	0.97	0.94	0.95	0.95
Mali		1.00	0.93	0.99	0.99	0.99	0.99	0.98
France			1.00	0.94	0.96	0.90	0.92	0.97
CI				1.00	0.99	0.99	0.99	0.98
Burkina Faso				1.00	0.98	0.99	0.99	
Benin						1.00	0.99	0.97
Togo							1.00	0.98
Senegal								1.00

Chi²(9) = 7.46

ANALYSIS OF THE DETERMINANTS OF FOREIGN DIRECT INVESTMENT FLOWS TO THE WAEMU COUNTRIES 31

Other recent publications in the AERC Research Papers Series:

An Empirical Analysis of the Determinants of Food Imports in Congo, by Léonard Nkouka Safoulanitou and Mathias Marie Adrien Ndinga, Research Paper 195.

Determinants of a Firm's Level of Exports: Evidence from Manufacturing Firms in Uganda, by Aggrey Niringiye and Richard Tuyiragize, Research Paper 196.

Supply Response, Risk and Institutional Change in Nigerian Agriculture, by Joshua Olusegun Ajetomobi, Research Paper 197.

Multidimensional Spatial Poverty Comparisons in Cameroon, by Aloysius Mom Njong, Research Paper 198.

Earnings and Employment Sector Choice in Kenya, by Robert Kivuti Nyaga, Research Paper 199.

Convergence and Economic Integration in Africa: the Case of the Franc Zone Countries, by Latif A.G. Dramani, Research Paper 200.

Analysis of Health Care Utilization in Côte d'Ivoire, by Alimatou Cisse, Research Paper 201.

Financial Sector Liberalization and Productivity Change in Uganda's Commercial Banking Sector, by Kenneth Alpha Egesa, Research Paper 202.

Competition and performace in Uganda's Banking System, by Adam Mugume, Research Paper 203.

Parallel Market Exchange Premiums and Customs and Excise Revenue in Nigeria, by Olumide S. Ayodele and Frances N. Obafemi, Research Paper 204.

Fiscal Reforms and Income Inequality in Senegal and Burkina Faso: A Comparative Study, by Mbaye Diene, Research Paper 205.

Factors Influencing Technical Efficiencies among Selected Wheat Farmers in Uasin Gishu District, Kenya, by James Njeru, Research Paper 206.

Exact Configuration of Poverty, Inequality and Polarization Trends in the Distribution of well-being in Cameroon, by Francis Menjo Baye, Research Paper 207.

- Child Labour and Poverty Linkages: A Micro Analysis from Rural Malawian Data, by Levision S. Chiwaula, Research Paper 208.
- The Determinants of Private Investment in Benin: A Panel Data Analysis, by Sosthène Ulrich Gnansounou, Research Paper 209.

Contingent Valuation in Community-Based Project Planning: The Case of Lake Bamendjim Fishery Restocking in Cameroon, by William M. Fonta et al., Research Paper 210.

- Multidimensional Poverty in Cameroon: Determinants and Spatial Distribution, by Paul Ningaye et al., Research Paper 211.
- What Drives Private Saving in Nigeria, by Tochukwu E. Nwachukwu and Peter Odigie, Research Paper 212.
- *Board Independence and Firm Financial Performance: Evidence from Nigeria*, by Ahmadu U. Sanda et al., Research Paper 213.
- *Quality and Demand for Health Care in Rural Uganda: Evidence from 2002/03 Household Survey,* by Darlison Kaija and Paul Okiira Okwi, Research Paper 214.

Capital Flight and its Determinants in the Franc Zone, by Ameth Saloum Ndiaye, Research Paper 215.

The Efficacy of Foreign Exchange Market Intervention in Malawi, by Kisukyabo Simwaka and Leslie Mkandawire, Research Paper 216.

The Determinants of Child Schooling in Nigeria, by Olanrewaju Olaniyan, Research Paper 217.

Influence of the Fiscal System on Income Distribution in Regions and Small Areas: Microsimulated CGE Model for Côte d'Ivoire, by Bédia F. Aka and Souleymane S. Diallo, Research Paper 218.

Asset Price Developments in an Emerging Stock Market: The Case of Mauritius, by Sunil K. Bundoo, Research Paper 219.

Intrahousehold Resource Allocation in Kenya, by Miriam Omolo, Research Paper 220.

- Volatility of Resource Inflows and Domestic Investment in Cameroon, by Sunday A. Khan, Research Paper 221.
- *Efficiency Wage, Rent-sharing Theories and Wage Determination in the Manufacturing Sector in Nigeria,* by Ben E. Aigbokhan, Research Paper 222.
- Government Wage Review Policy and Public-Private Sector Wage Differential in Nigeria, by Alarudeen Aminu, Research Paper 223.
- Rural Non-Farm Incomes and Poverty Reduction in Nigeria, by Awoyemi Taiwo Timothy, Research Paper 224.

- What Do We Know After 15 Years of Using the United Nations Development Programme Human Development Index? By Jean Claude Saha, Research Paper 225.
- Uncertainty and Investment Behaviour in the Democratic Republic of Congo, by Xavier Bitemo Ndiwulu and Jean-Papy Manika Manzongani, Research Paper 226.
- An Analysis of Stock Market Anomalies and Momentum Strategies on the Stock Exchange of Mauritius, by S.K. Bundoo, Research Paper 227.
- The Effect of Price Stability on Real Sector Performance in Ghana, by Peter Quartey, Research Paper 228.
- The Impact of Property Land Rights on the Production of Paddy Rice in the Tillabéry, Niamey and Dosso Regions of Niger, by Maman Nafiou Malam Maman and Boubacar Soumana, Research Paper 229.
- An Econometric Analysis of the Monetary Policy Reaction Function in Nigeria, by Chukwuma Agu, Research Paper 230.
- Investment in Technology and Export Potential of Firms in Southwest Nigeria, by John Olatunji Adeoti, Research Paper 231.
- Analysis of Technical Efficiency Differentials among Maize Farmers in Nigeria, by Luke Oyesola Olarinde, Research Paper 232.
- Import Demand in Ghana: Structure, Behaviour and Stability, by Simon Harvey and Kordzo Sedegah, Research Paper 233.
- Trade Liberalization Financing and its Impact on Poverty and Income Distribution in Ghana, by Vijay K. Bhasin, Research Paper 234.
- An Empirical Evaluation of Trade Potential in Southern African Development Community, by Kisukyabo Simwaka, Research Paper 235.
- Government Capital Spending and Financing and its Impact on Private Investment in Kenya: 1964-2006, by Samuel O. Oyieke, Research Paper 236.
- Determinants of Venture Capital in Africa: Cross Section Evidence, by Jonathan Adongo, Research Paper 237.
- Social Capital and Household Welfare in Cameroon: A Multidimensional Analysis, by Tabi Atemnkeng Johannes, Research Paper 238.